Author's response to reviews

Title: Contribution of chronic diseases to the disability burden in Belgium

Authors:

Renata TC Yokota (renata.yokota@wiv-isp.be)
Nicolas Berger (nicolas.berger@wiv-isp.be)
Wilma J Nusselder (w.nusselder@erasmusmc.nl)
Jean-Marie Robine (jean-marie.robine@inserm.fr)
Jean Tafforeau (jean.tafforeau@wiv-isp.be)
Patrick Deboosere (patrick.deboosere@vub.ac.be)
Herman Van Oyen (Herman.VanOyen@wiv-isp.be)

Version: 3
Date: 16 September 2014

Author's response to reviews: see over
Dear Editors of BMC Public Health,

Thank you for considering our manuscript for peer review. We are re-submitting the article with the required information included. Please find the reply to your specific request in bold below.

1.) Please state in your manuscript whether the data used is openly available or whether you received permission to use this (and from who).

The Belgian HIS data is not public available. Under a contract with the Statistics Belgium, that is responsible for the HIS data, all employees of the Scientific Institute of Public Health in Belgium are allowed to do the data analysis, according to the ethical procedures of the Institute.

This information was added to the paper in page 4, lines 15-17, in red.

We are looking forward to the reviewers’ reply.

Kind regards,

Renata Yokota